Author's response to reviews

Title: Anticancer activity of extracts derived from the mature roots of Scutellaria baicalensis on human malignant brain tumor cells

Authors:

Adrienne C Scheck (adrienne.scheck@chw.edu)
Krya Perry (krya.perry@chw.edu)
Nicole C. Hank (nicole.hank@chw.edu)
W. DENNIS Clark (wdclark@usbotanicals.net)

Version: 4 Date: 15 August 2006

Author's response to reviews:

We have made the requested format changes